Author's response to reviews

Title: Arm Rotated Medially with Supination (the ARMS Deformity): Description of a New Clinical Entity and its Surgical Correction

Authors:

Rahul K Nath (drnath@drnathmedical.com)
Chandra Somasundaram (chandra@drnathmedical.com)
Sonya E Melcher (sonya@drnathmedical.com)
Meera Bala (meera@drnathmedical.com)
Melissa J Wentz (melissa@drnathmedical.com)

Version: 2 Date: 14 May 2008

Author's response to reviews:

Dear Dr. Marlee,

We have revised the abstract, improving it and including the four sections as requested. We have also added a competing interests section and formatted the manuscript according to the BMC template. If there were additional changes requested, please contact us.

Thank you for considering our manuscript.

Dr RK Nath